\begin{table}%T1
\par
\caption{\label{tab:datatts}Orientation of T~Tauri stars in Taurus-Auriga.}
%\centerline
{
\begin{tabular}{lllccccccc}
\hline\hline
Object & $\alpha (2000)$ & $\delta (2000)$ & PA (jet) &
Ref. & PA (disk) & Ref. & PA ($\vec{B}$) &
$\rho_{\rm IS}$/$N_{\rm IS}$ & $|\Delta$PA$|$ \\
 & & & \degr &  & \degr & & \degr & & \degr \\
(1) & (2)& (3) & (4) & (5) & (6) & (7) & (8) & (9) & (10) \\
\hline
CW Tau & 04$^{\rm h}$14$^{\rm m}$16\fs95 &
$28\degr10\arcmin59\farcs3$ & 144~$\pm$~2 & 1 & -- & -- & 15 &
1.5/12 & 51 \\
DD Tau A & 04 18 31.13 & 28 16 30.1 & 125 & 26 & -- & -- & 15 & 0.5 /
5 & 70 \\
CoKu Tau/1 & 04 18 51.54 & 28 20 28.1 & 28 & 11 & 120 & 4
& 15 & 0.5/5 & 13 \\
04158+2805 & 04 18 58.2 & 28 12 23 & 165 & 5 & 89 &
5 & 15 & 0.5/5 & 30 \\
RY Tau & 04 21 57.34 & 28 26 36.9 & 110 & 14 &
31~$\pm$~6 & 6 & 18 & 0.5/10 & 88 \\ 
T Tau N & 04 21 59.39 & 19 32 06.8 & 90 & 18 & 19 &
19 & 85 & 1.5/4 & 5 \\ 
T Tau S & 04 21 59.39 & 19 32 06.1 & 170 & 18 & -- &
-- & 85 & 1.5/4 & 85 \\
Haro 6-5B & 04 22 00.89 & 26 57 37.6 & 53 & 7 &
152~$\pm$~2 & 6 & 22 & 0.5/32 & 31 \\ 
FT Tau & 04 23 39.17 & 24 56 15.1 & 29 & 14 &
82~$\pm$~17 & 8 & 73 & 0.5/18 & 44 \\
DG Tau B & 04 27 02.56 & 26 05 30.7 & 111~$\pm$~8 & 11
& 32 & 4 & 32 & 0.5/16 & 79 \\
DF Tau & 04 27 02.80 & 25 42 22.3 & 127 & 26 & -- & -- & 35 & 1.0/44
& 88 \\
DG Tau & 04 27 04.67 & 26 06 16.9 & 42 & 9 & 136 &
10 & 30 & 0.5/18 & 12 \\
Haro 6-10 & 04 29 23.65 & 24 33 02.0 & 60 & 12 & -- & --
& 74 & 0.5/12 & 14 \\
HH 30 & 04 31 37.6 & 18 12 26 & 31 & 23 & 121 & 23 &
77 & 0.5/13 & 46 \\
HL Tau & 04 31 38.44 & 18 13 58.9 & 51 & 7 &
125~$\pm$~10 & 20 & 77 & 0.5/13 & 26 \\
XZ Tau & 04 31 40.02 & 18 13 57.8 & 15 & 21 & -- & -- &
77 & 0.5/13 & 62 \\
Haro~6-13 & 04 32 15.61 & 24 29 02.3 & 65 & 25 & -- & -- & 
66 & 0.5/16 & 1 \\
UZ Tau E & 04 32 42.93 & 25 52 32.1 & 0 & 12 &
86~$\pm$~2 & 2 & 30 & 1.0/16 & 30 \\
GK Tau & 04 33 34.47 & 24 21 07.6 & 20 & 24 & -- & -- & 50 &
0.5/19 & 30 \\
DL Tau & 04 33 39.04 & 25 20 38.9 & 145 & 14 &
50~$\pm$~3 & 2 & 53 & 0.5/4 & 88 \\
HN Tau & 04 33 39.32 & 17 51 52.8 & 170 & 12 & -- & -- &
77 & 1.0/13 & 87 \\
DO Tau & 04 38 28.57 & 26 10 50.5 & 70~$\pm$~10 & 12 &
61~$\pm$~5 & 6 & 56 & 0.5/7 & 14 \\
HV Tau C & 04 38 35.29 & 26 10 40.0 & 24 & 14 &
110 & 15 & 56 & 0.5/7 & 32 \\
Haro 6-33 & 04 41 38.9 & 25 56 26 & 60 & 14 & -- & -- & 52 &
0.5/9 & 8 \\
DP Tau & 04 42 37.66 & 25 15 38.1 & 40 & 16 & -- & -- &
39 & 0.5/5 & 1 \\
UY Aur & 04 51 47.36 & 30 47 14.1 & 40 & 12 &
135~$\pm$~5 & 17 & 65 & 2.0/8 & 25 \\
\hline
CY Tau & 04 17 33.73 & 28 20 47.8 & -- & -- &
150~$\pm$~7 & 2 & 15 & 1.0/8 & 45 \\ 
BP Tau & 04 19 15.82 & 29 06 27.9 & -- & -- &
152~$\pm$~3 & 2 & 175 & 0.5/5 & 67 \\ 
IQ Tau & 04 29 51.50 & 26 06 46.5 & -- & -- &
29~$\pm$~2 & 6 & 26 & 0.5/12 & 87 \\
HK Tau B & 04 31 50.55 & 24 24 18.4 & -- & -- & 40 & 13 &
70 & 0.5/14 & 60 \\
GG Tau & 04 32 30.27 & 17 31 41.5 & -- & -- &
97~$\pm$~2 & 22 & 77 & 1.0/13 & 70 \\
DM Tau & 04 33 48.70 & 18 10 10.7 & -- & -- &
153~$\pm$~1 & 2 & 75 & 0.5/8 & 12 \\
CI Tau & 04 33 51.99 & 22 50 30.6 & -- & -- &
40~$\pm$~18 & 8 & 54 & 1.0/4 & 76 \\
AA Tau & 04 34 55.40 & 24 28 53.8 & -- & -- &
75~$\pm$~16 & 6 & 51 & 0.5/21 & 65 \\
DN Tau & 04 35 27.35 & 24 14 59.8 & -- & -- &
120~$\pm$~5 & 6 & 52 & 0.5/22 & 22 \\
LkCa 15 & 04 39 17.80 & 22 21 04.5 & -- & -- &
61~$\pm$~1 & 2 & 54 & 1.5/4 & 83 \\
GM Aur & 04 55 10.95 & 30 22 01.0 & -- & -- &
51~$\pm$~2 & 2 & 65 & 2.0/9 & 76 \\
\hline
\end{tabular}}
\smallskip
References: 1)~Dougados et~al. (2000), 2)~Simon et~al. (\cite{Simon2000}),
3)~Gomez~de~Castro \& Pudritz (\cite{Gomez1992}), 4)~Padgett et~al. (\cite{Padgett1999}), 5)~M\'enard et~al. (\cite{enard2004}), 6)~Kitamura et~al. (\cite{Kitamura2002}), 7)~Mundt et~al. (\cite{Mundt1991}), 8)~Dutrey et~al. (\cite{Dutrey1996}),
9)~Lavalley et~al. (\cite{Lavalley1997}), 10)~Kitamura et~al.
(\cite{Kitamura1996}), 11)~Eisl\"offel \& Mundt (1998), 12)~Hirth et~al. (\cite{Hirth1997}), 13)~Stapelfeldt et~al. (\cite{Stapelfeldt1998}), 14)~Stapelfeldt et~al. (2004, in prep.), 15)~Monin \& Bouvier (\cite{Monin2000}), 16)~Mundt
\& Eisl\"offel (\cite{Mundt1998}), 17)~Close et~al. (\cite{Close1998}), 18)~Solf \&
B\"ohm (\cite{Solf1999}), 19)~Akeson et~al. (\cite{Akeson1998}), 20)
Wilner et~al.  (\cite{Wilner1996}), 21)~Mundt et~al. (1990), 22)
Guilloteau et~al.  (\cite{Guilloteau1999}), 23)~Burrows et~al. (\cite{Burrows1996}),
24)~Aspin \& Reipurth (\cite{Aspin2000}), 25)~Strom et~al. (\cite{Strom1986}), 26)~Hartigan et~al. (\cite{Hartigan2004}).
\end{table}